Author's response to reviews

Title: A study of diverse clinical decision support rule authoring environments and requirements for integration

Authors:

Li Zhou (lzhou2@partners.org)
Neelima Karipineni (nkaripineni@partners.org)
Janet Lewis (jlewis18@partners.org)
Saverio M Maviglia (smaviglia@partners.org)
Amanda Fairbanks (afairbanks@partners.org)
Blackford Middleton (bmiddleton1@partners.org)
Roberto A Rocha (rarocha@partners.org)

Version: 4 Date: 2 September 2012

Author's response to reviews: see over
September 2, 2012

BMC Medical Informatics & Decision Making

Dear Editors,

We are resubmitting our revised manuscript, entitled “A study of diverse clinical decision support rule authoring environments and requirements for integration” (MS: 4911017107484), for possible publication in the BMC Medical Informatics & Decision Making. This manuscript has not been submitted for publication nor has it been published in whole or in part elsewhere.

Thank you for your review of our manuscript. We appreciate the four reviewers’ valuable comments and revised the manuscript based on their suggestions. We addressed each specific item of each reviewer’s comments, point-by-point. We also mentioned the corresponding changes made in the revised text.

Thank you very much for considering our work. We look forward to your review of our revised manuscript.

Sincerely,

Li Zhou